Author’s response to reviews

Title: Images of God and attitudes towards death in relation to spiritual wellbeing: an exploratory side study of the EORTC QLQ-SWB32 validation study in palliative cancer patients

Authors:

Renske Kruizinga (r.kruizinga@amc.uva.nl)
Michael Scherer-Rath (m.scherer-rath@ftr.ru.nl)
Johannes Schilderman (h.schilderman@ftr.ru.nl)
Mariëtte Weterman (m.j.weterman@amc.uva.nl)
Teresa Young (teresa.young2@nhs.net)
Hanneke van Laarhoven (h.vanlaarhoven@amc.uva.nl)

Version: 2 Date: 17 Nov 2017

Author’s response to reviews:

Dear Editor,

Please find uploaded our revised manuscript “Images of God and attitudes towards death in relation to spiritual wellbeing: an exploratory side study of the EORTC QLQ-SWB32 validation study in palliative cancer patients” to be reconsidered for publication.

We are very glad to have the opportunity to submit a revised version of our paper for publication in BMC Palliative Care. We thank the editor and referees for the thoughtful reading of the manuscript and the supportively critical comments. These were very helpful and we believe that the changes have improved our manuscript.

All author details on the revised version are correct, all authors have agreed to authorship and order of authorship for this manuscript and all authors have the appropriate permissions and rights to the reported data.

Thank you again for considering our manuscript for publication.

Yours sincerely,

Drs. Renske Kruizinga,

Also on behalf of my co-authors,
Michael Scherer-Rath
Johannes B.A.M. Schilderman
Mariëtte Weterman
Teresa Young
Hanneke W.M. van Laarhoven